6.891: Questionnaire

• Your name:

• Your email address:

• Which program are you in? (Undergraduate, MENG, or PhD; in computer science, linguistics, or another field; at MIT or elsewhere?)

• Have you taken classes in the past which are related to this course?

(in particular, previous courses in AI/NLP/Linguistics/Machine Learning/speech recognition?)

• Do you have any background in probability and/or statistics?

• Have you registered for the class (or are you intending to register for the class)?

• Do you speak any languages other than English?

• Any other comments?